Author's response to reviews

Title: HIV, HSV-2 and syphilis among pregnant women in a rural area of Tanzania: Prevalence and risk factors

Authors:

Khadija I Yahya-Malima (khadija.malima@cih.uib.no)
Bjorg Evjen-Olsen (bjorg.olsen@cih.uib.no)
Mecky I Matee (mateerrr@yahoo.com)
Knut Fylkesnes (knut.fylkesnes@cih.uib.no)
Lars Haarr (lars.haarr@vir.uib.no)

Version: 7 Date: 12 May 2008

Author's response to reviews: see over
Dear Sir/ Madam

Re: Review and resubmission of a revised manuscript MS: 1991897168153309 renamed as “HIV, HSV-2 and syphilis among pregnant women in a rural area of Tanzania: Prevalence and risk factors”

We would like to thank you for accepting the manuscript for publication. We have now presented the background in the abstract as required and the word count in the abstract is now 289 words. We hope that you will find the manuscript is now satisfactory.

Sincerely

Khadija I. Yahya-Malima

Corresponding author: Khadija I. Yahya-Malima
Centre for International Health
Armour Hansen Building
University of Bergen
Bergen, Norway
E-mail: khadija.malima@cih.uib.no
Phone: +47 55 97 49 80/ +255 784 627 380
Fax: +47 55 97 49 79